Author's response to reviews

Title: Tuberculosis case finding based on symptom screening among immigrants, refugees and asylum seekers in Rome

Authors:

Monica Sañé Schepisi (monica.saneschepisi@inmi.it)
Gina Gualano (gualano@inmi.it)
Claudia Fellus (CLAUDIAFELLUS@ALICE.IT)
Nazario Bevilacqua (bevilacqua@inmi.it)
Marco Vecchi (mv7110@gmail.com)
Pierluca Piselli (pierluca.piselli@inmi.it)
Giuliana Battagin (batgin@yahoo.com)
Giulia Silvestrini (silvestrinigiulia@gmail.com)
Andrea Attanasio (andreattanasi@hotmail.com)
Alberto Vela (A.Vela@aslromah.it)
Giorgia Rocca (giorgia_rocca@hotmail.com)
Alessandro Rinaldi (a.rinaldi28@virgilio.it)
Pietro Benedetti (pbenedetti74@gmail.com)
Salvatore Geraci (s.geraci@areasanitaria.it)
Francesco N Lauria (lauria@inmi.it)
Enrico Girardi (enrico.girardi@inmi.it)

Version: 4 Date: 11 September 2013

Author's response to reviews: see over
Dear Editor,

Thanks for your email regarding our manuscript entitled: “Tuberculosis case finding based on symptom screening among immigrants, refugees and asylum seekers in Rome” (MS: 8696677968937012)

We did appreciate the reviewers’ comments and we modified the manuscript accordingly as detailed below.

We hope you will find this revised version suitable for publication in BMC Public Health

With best regards
Enrico Girardi, MD

Reviewer 1
- I am satisfied by the response to the previous review.
We thank the reviewer for this statement

Discretional revision:
- 1: line 136-137: "47,2% of the subjects had spent in Italy less than 2 years" - could be changed to "47,2% of the subjects had spent less than 2 years in Italy"
- 2: line 212-213: "originating for Asia" changed to: "originating from Asia"
- 3: line 228-229: "Thus, taken together, available--" changed to: "Thus, available"
We corrected the manuscripts as suggested by the reviewer

Reviewer 2
- The authors have provided enough information for addressing my comments. I don’t have further comments on the revision.
We thank the reviewer for this statement